## **PLOS MEDICINE**



## 

**Citation:** Rogne T, Solligård E, Burgess S, Brumpton BM, Paulsen J, Prescott HC, et al. (2020) Body mass index and risk of dying from a bloodstream infection: A Mendelian randomization study. PLoS Med 17(11): e1003413. https://doi. org/10.1371/journal.pmed.1003413

Academic Editor: Cosetta Minelli, Imperial College London, UNITED KINGDOM

Received: December 18, 2019

Accepted: October 8, 2020

Published: November 16, 2020

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003413

**Copyright:** © 2020 Rogne et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data from the HUNT Study used in research projects available upon request to the HUNT Data Access Committee (hunt@medisin.ntnu.no) to research groups who RESEARCH ARTICLE

# Body mass index and risk of dying from a bloodstream infection: A Mendelian randomization study

Tormod Rogne<sup>1,2,3</sup>\*, Erik Solligård<sup>1,3</sup>, Stephen Burgess<sup>4,5</sup>, Ben M. Brumpton<sup>6,7,8</sup>, Julie Paulsen<sup>9</sup>, Hallie C. Prescotto<sup>10,11</sup>, Randi M. Mohus<sup>1,3</sup>, Lise T. Gustad<sup>1,12</sup>, Arne Mehl<sup>12</sup>, Bjørn O. Åsvold<sup>6,13</sup>, Andrew T. DeWan<sup>1,2‡</sup>, Jan K. Damås<sup>1,14,15‡</sup>

1 Gemini Center for Sepsis Research, Department of Circulation and Medical Imaging, NTNU Norwegian University of Science and Technology, Trondheim, Norway, 2 Department of Chronic Disease Epidemiology, Yale University School of Public Health, New Haven, Connecticut, United States of America, 3 Clinic of Anaesthesia and Intensive Care, St. Olavs Hospital, Trondheim University Hospital, Trondheim, Norway, 4 MRC Biostatistics Unit, University of Cambridge, Cambridge, United Kingdom, 5 Cardiovascular Epidemiology Unit, Department of Public Health and Primary Care, University of Cambridge, Cambridge, United Kingdom, 6 K.G. Jebsen Center for Genetic Epidemiology, Department of Public Health and Nursing, NTNU Norwegian University of Science and Technology, Trondheim, Norway, 7 MRC Integrative Epidemiology Unit, University of Bristol, Bristol, United Kingdom, 8 Clinic of Thoracic and Occupational Medicine, St. Olavs Hospital, Trondheim University Hospital, Trondheim, Norway, 9 Department of Medical Genetics, St. Olavs Hospital, Trondheim University Hospital, Trondheim, Norway, 10 Department of Medicine, University of Michigan, Ann Arbor, Michigan, United States of America, 11 VA Center for Clinical Management Research, Ann Arbor, Michigan, United States of America, 12 Department of Medicine, Levanger Hospital, Nord-Trøndelag Hospital Trust, Levanger, Norway, 13 Department of Endocrinology, St. Olavs Hospital, Trondheim University Hospital, Trondheim, Norway, 14 Centre of Molecular Inflammation Research, Department of Clinical and Molecular Medicine, NTNU Norwegian University of Science and Technology, Trondheim, Norway, 15 Department of Infectious Diseases, St. Olavs Hospital, Trondheim University Hospital, Trondheim, Norway

‡ These authors are joint senior authors on this work. \* tormod.rogne@yale.edu

### Abstract

#### Background

In observational studies of the general population, higher body mass index (BMI) has been associated with increased incidence of and mortality from bloodstream infection (BSI) and sepsis. On the other hand, higher BMI has been observed to be apparently protective among patients with infection and sepsis. We aimed to evaluate the causal association of BMI with risk of and mortality from BSI.

#### Methods and findings

We used a population-based cohort in Norway followed from 1995 to 2017 (the Trøndelag Health Study [HUNT]), and carried out linear and nonlinear Mendelian randomization analyses. Among 55,908 participants, the mean age at enrollment was 48.3 years, 26,324 (47.1%) were men, and mean BMI was 26.3 kg/m<sup>2</sup>. During a median 21 years of follow-up, 2,547 (4.6%) participants experienced a BSI, and 451 (0.8%) died from BSI. Compared with a genetically predicted BMI of 25 kg/m<sup>2</sup>, a genetically predicted BMI of 30 kg/m<sup>2</sup> was associated with a hazard ratio for BSI incidence of 1.78 (95% CI: 1.40 to 2.27; p < 0.001) and for

meet the data availability requirements (described here: http://www.ntnu.edu/hunt/data). BMI GWAS source data (Yengo et al, 2018) are available from the GIANT Consortium (https://portals. broadinstitute.org/collaboration/giant/index.php/ GIANT\_consortium\_data\_files).

Funding: This study was in part funded by Samarbeidsorganet Helse Midt-Norge, NTNU, and The Research Council of Norway. TR was funded in part by a Fulbright Scholarship by the U.S-Norway Fulbright Foundation. BMB and BOÅ work in a research unit funded by Stiftelsen Kristian Gerhard Jebsen; Faculty of Medicine and Health Sciences, NTNU; The Liaison Committee for education, research and innovation in Central Norway; the Joint Research Committee between St. Olavs Hospital and the Faculty of Medicine and Health Sciences, NTNU; and the Medical Research Council Integrative Epidemiology Unit at the University of Bristol which is supported by the Medical Research Council and the University of Bristol [MC\_UU\_12013/1]. HCP is funded in part by K08 GM115859 from the US National Institutes of Health. The funding sources had no role in study design; in the collection, analysis, and interpretation of data; in the writing of the report; nor in the decision to submit the article for publication. The researchers were independent from the funders, and all authors had full access to all of the data in the study and can take responsibility for the integrity of the data and the accuracy of the data analysis.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: SB is a paid statistical reviewer for PLOS Medicine. HCP has current or prior grant funding from the NIH, AHRQ, and the US Department of Veterans Affairs. The other authors have declared that no competing interests exist.

**Abbreviations:** BMI, body mass index; BSI, bloodstream infection; GRS, genetic risk score; HR, hazard ratio; HUNT, Trøndelag Health Study; IPW, inverse probability weighting; IVW, inversevariance-weighted; MR, Mendelian randomization. BSI mortality of 2.56 (95% CI: 1.31 to 4.99; p = 0.006) in the general population, and a hazard ratio for BSI mortality of 2.34 (95% CI: 1.11 to 4.94; p = 0.025) in an inverse-probabilityweighted analysis of patients with BSI. Limitations of this study include a risk of pleiotropic effects that may affect causal inference, and that only participants of European ancestry were considered.

#### Conclusions

Supportive of a causal relationship, genetically predicted BMI was positively associated with BSI incidence and mortality in this cohort. Our findings contradict the "obesity paradox," where previous traditional epidemiological studies have found increased BMI to be apparently protective in terms of mortality for patients with BSI or sepsis.

#### Author summary

#### Why was this study done?

- It is well-recognized that overweight and obesity are associated with increased risk of bloodstream infection (BSI) and sepsis, but it is not fully understood whether this is due to body weight in itself or factors related to body weight (such as exercise or smoking habits).
- While a large number of studies have observed that BSI or sepsis patients who are overweight or obese have a *reduced* risk of dying from those diseases, there is reason to suspect that these findings are biased.
- We wanted to evaluate whether genetically predicted body mass index (BMI)—which is independent of lifestyle factors—was associated with risk of developing and dying from a BSI.

#### What did the researchers do and find?

- We used clinical and genetic information from the Trøndelag Health Study in Norway on 55,908 participants representative of the adult Norwegian population.
- Similar to what has been found in non-genetic studies, we found that increased genetically predicted BMI was associated with an increased risk of developing a BSI.
- Contrary to many observational studies, we found that among BSI patients, being overweight or obese was associated with an increased risk of death from bloodstream infection.

#### What do these findings mean?

• The findings of many previous observational studies of an apparently protective effect of overweight or obesity among patients with BSI or sepsis may be affected by other factors, such as accompanying characteristics of overweight or obese individuals, or by who ends up participating in the studies. • In this cohort, higher genetically predicted BMI was associated with an increased risk of developing and dying from a BSI, also among BSI patients, and our findings support the worldwide initiative to reduce the prevalence of overweight and obesity.

#### Introduction

Bloodstream infection (BSI) is caused by bacteria entering the bloodstream as a severe complication of an infection, and may in turn lead to sepsis, representing a dysregulated immune response to infection resulting in organ dysfunction and high mortality rate [1]. BSI is a common cause of death globally [2–4]. It is therefore important to identify factors that may reduce the risk of developing and dying from BSI.

Obesity is increasingly common worldwide [5], and is associated with greater risk of a wide range of diseases and all-cause mortality [6]. Traditional observational studies of general populations have found body mass index (BMI) to be positively associated with risk of BSI or sepsis [7–10], and above-normal BMI to be associated with BSI or sepsis mortality [7,9,11]. However, studies restricted to patients with BSI or sepsis have observed a considerable *reduced* mortality risk with increasing BMI [12–17], including a systematic review from 2017 [18]. It has therefore been advocated that one should study in what way obesity is protective for sepsis mortality, and that this may help inform new therapeutic strategies [17,18]. This counterintuitive reduction in disease progression or mortality among overweight and obese patients with a particular disease has been termed the "obesity paradox" and has been observed for other illnesses such as type 2 diabetes [19].

Some argue that the paradox may be explained by systematic error, and there has been a call for more rigorous studies to establish causal relationships [20–22]. There are 3 main areas of concern: selection bias, reverse causation, and confounding. If obesity is associated with BSI risk, nonobese patients may have other characteristics that caused their BSI that in turn are more strongly associated with mortality. This selection bias may make obesity appear protective in studies of mortality rate among all patients with BSI or sepsis [23]. Reverse causation may arise if measured BMI is affected by BSI (e.g., dehydration). Importantly, there may be confounding from factors such as chronic diseases and smoking habits that affect both BMI and BSI mortality. Mendelian randomization (MR) studies mimic randomized trials using genetic data as instruments for exposures. MR leverages information on genetic variants that segregate randomly at conception. If people with a genetic risk of being overweight or obese also have an increased risk of developing and dying from BSI, a causal relationship between BMI and mortality from BSI is strengthened (i.e., the relationship is likely independent of confounders and not subject to reverse causation) [24].

The aim of this study was to assess the causal association between BMI and risk of and mortality from BSI. We sought to overcome the limitations of the observational studies mentioned above by conducting an MR study in a general population of approximately 56,000 participants in Norway with 23 years of follow-up.

#### Methods

This study is reported according to the preprint guideline for reporting of MR studies, STRO-BE-MR (S1 Checklist) [25] and the STROBE guideline for cohort studies (S2 Checklist) [26]. The Trøndelag Health Study (HUNT) is a series of cross-sectional surveys carried out in Nord-Trøndelag County, Norway. The county consists of 130,000 inhabitants, and is representative of the general adult Norwegian population in terms of morbidity, mortality, sources of income, and age distribution [27]. The present study was based on the HUNT2 survey conducted in 1995–1997, to which 93,865 people were invited, and 65,236 (69.5%) participated.

Background characteristics such as age, sex, smoking status (current, former, never), lifestyle factors, education level, activity level, anthropometric measures, and self-reported history of cancer were collected once for each participant in the HUNT2 survey. Height and weight were measured by trained staff, and BMI calculated as weight in kilograms divided by the squared height in meters.

The personal identification number of Norwegian citizens was used to link the study population to all prospectively recorded blood cultures at the 2 community hospitals in the catchment area (Levanger and Namsos Hospitals), as well as St. Olavs Hospital in Trondheim, which serves as a tertiary referral center. Data on blood cultures were available from 1 January 1995 (from 1 September 1999 in Namsos Hospital) through the end of 2017. Dates of death and emigration out of Nord-Trøndelag County were obtained from the Norwegian population registry.

BSI was defined as positive blood culture of pathogenic bacteria, excluding bacteria such as *Corynebacterium* species, *Propionibacterium* species, and coagulase-negative *Staphylococcus* species often associated with blood culture contamination [28]. BSI mortality was defined as death within 30 days of BSI diagnosis.

Participants were followed until death or emigration out of Nord-Trøndelag, or to the end of December 2017.

Details about genotyping and imputation are provided in <u>S1 Text</u>. After sample and variant quality control, imputation was completed for 61,412 patients, all of European ancestry.

The genetic risk score (GRS) for BMI was calculated based on 939 of 941 near-independent, single nucleotide polymorphisms (SNPs;  $p < 1 \times 10^{-8}$ ) identified as related to BMI in a metaanalysis of approximately 700,000 individuals (rs498240 and rs3819299 did not pass imputation quality control) [29].

Among patients with genotype data, BMI was available for 55,944. One participant was excluded due to outlying weight/height ratio. Participants who had a registered BSI before participating in the HUNT2 survey were excluded (n = 35). The final study sample consisted of 55,908 participants.

No protocol was written for this particular study, but the aims of the study were formalized prior to the analyses being conducted. The only substantial changes to the analyses were the addition of 2-sample MR analyses as suggested by peer-reviewers.

#### Statistical analysis

The a priori main outcome of this MR study was the association between genetically predicted BMI and BSI mortality in the general population, while secondary analyses were MR analyses of association between genetically predicted BMI and (1) BSI incidence, (2) BSI incidence and mortality stratified by sex, and (3) BSI mortality among patients with BSI. Unless stated otherwise, the results presented in this study are evaluating the effect of genetically predicted BMI (as opposed to observed BMI).

The GRS for BMI was calculated by use of the "—score" command in PLINK (version 1.9), weighted based on the effect estimates from the meta-analysis by Yengo et al. [29]. We estimated the association between the GRS and BMI by use of linear regression, and between the GRS and risk of BSI or BSI mortality by use of Cox proportional hazards regression. Both models accounted for age and sex.

As previous studies have suggested a nonlinear association between BMI and BSI incidence and mortality [9], we fitted a fractional polynomial model as described elsewhere [30] and in S1 Text. In analyses assuming a linear relationship between exposure and outcome, we used the 2-stage least squares method with sandwich estimator for the error terms in the second stage [31]. In the MR analysis of the effect of BMI on BSI mortality among patients with BSI, day 0 was set to day of BSI diagnosis, and participants were followed for 30 days. To account for potential selection bias, we weighted participants from the total study population by the inverse probability of developing BSI, calculated based on the GRS, BMI, age, sex, history of cancer, smoking status, physical activity, and education [32]. The weighting was trimmed at the 99th percentile. This analysis was compared to an MR analysis without inverse probability weighting (IPW) and a similar traditional multivariable analysis adjusting for age, sex, history of cancer, smoking status, physical activity, and education.

Three main assumptions must be met for an MR study to be valid: (1) There is an association between the genetic instrument and the exposure; (2) the genetic instrument is not associated with the outcome other than through the exposure; and (3) the genetic instrument is not associated with confounders of the exposure-outcome association [24]. We performed a range of sensitivity analyses: (1) an analysis of the linear association between the GRS and BMI; (2) an analysis of the association between quartiles of the GRS and predefined confounders (due to signs of pleiotropic effects on smoking habits, the main analysis was run among neversmokers and then among previous or current smokers); (3) an MR analysis of BSI mortality in the general population restricted to near-independent SNPs associated with the FTO gene (robustly associated with BMI) [33]; (4) MR-Egger regression (random effects), inverse-variance-weighted (IVW) regression (random effects), and weighted median estimator analysis [34], assuming a linear relationship between BMI and BSI mortality in the general population; (5) comparison of analyses from the previous point with the same analyses conducted in a 2-sample design, using SNP-exposure associations from Yengo et al. [29] and SNP-outcome associations from HUNT (methods described in S1 Text); (6) comparison of our main MR analysis of BSI mortality in the general population with comparable traditional Cox regression adjusting for age, sex, history of cancer, smoking, physical activity, and education; and (7) repetition of the main analysis after adjustment for 5 ancestry-informative principal components computed by use of LASER v2.04 to account for potential population stratification [35].

Signs of selection bias were explored by restricting the study population to participants with BSI. We compared values of confounders for individuals in the highest BMI quartile (Q4) to those for individuals in the lowest BMI quartile (Q1) among patients with BSI, and compared this ratio (Q4/Q1) or difference (Q4 - Q1) with the concordant comparison in the total study population.

Finally, we did an MR sensitivity analysis stratified on whether the BSI was due to grampositive or gram-negative bacteria.

R (version 3.4.1; packages metafor, ggplot2, gtools, Hmisc, ipw, lm.beta, sandwich, survey, survival, timereg, and lmtest) and Stata/SE 15.1 (College Station, TX, US; package mrrobust) were used.

#### **Ethical approval**

The Regional Committee for Medical Research, Health Region IV, in Norway approved HUNT, and this project is regulated in conjunction with Norwegian Social Science Data Services.

#### Results

A total of 55,908 participants were followed for a median of 21 years, amounting to 995,474 person-years. There were 2,547 (4.6%) first occurrences of BSI, and 451 (0.8% of total, 17.7%

Characteristic	Total population $(n = 55.908)$	BSI incidence $(n = 2.547)$	BSI death $(n = 451)$
Age (years) <sup>§</sup>	48.3 (36.5-62.3)	63.6 (52.9–71.4)	67.3 (57.1–74.5)
Male sex*	26,324 (47.1)	1,345 (52.8)	263 (58.3)
BMI (kg/m <sup>2</sup> )^	26.3 (4.1)	27.7 (4.5)	27.9 (4.8)
Median follow-up time (years) <sup>§</sup>	21.1 (17.1–21.8)	13.8 (8.4–18.3)	13.3 (7.7–17.9)
Self-reported cancer*	1,955 (3.7)	144 (6.2)	24 (5.9)
Smoking*			
Never	23,594 (43.0)	876 (35.2)	156 (35.6)
Previous	15,133 (27.6)	893 (35.8)	164 (37.4)
Current	16,117 (29.4)	723 (29.0)	118 (26.9)
Physical activity*			
None	3,821 (7.6)	243 (11.9)	54 (15.4)
Slight	15,662 (31.0)	714 (34.9)	117 (33.3)
Moderate	17,167 (34.0)	693 (33.9)	116 (33.1)
High	13,810 (27.4)	397 (19.4)	64 (18.2)
Education*			
$\leq$ 9 years	19,033 (35.7)	1,305 (55.8)	240 (58.8)
10–12 years	23,468 (44.0)	762 (32.6)	125 (30.6)
$\geq$ 13 years	10,832 (20.3)	274 (11.7)	43 (10.5)

#### Table 1. Background characteristics.

BMI, body mass index; BSI, bloodstream infection. Data are presented as

^mean (standard deviation)

<sup>§</sup>median (25th–75th percentiles), or

\**n* (%). BSI incidence is based on first occurrence; otherwise, last occurrence is used. Education defined as follows:  $\leq 9$  years ("primary school 7–10 years, continuation school, folk high school"), 10–12 years ("high school, intermediate school, vocational school, 1–2 years high school" and "university qualifying examination, junior college, A levels"), and  $\geq 13$  years ("university or other post-secondary education, less than 4 years" and "university/college 4 years or more"). Activity defined as follows: none ("no light or vigorous activity"), slight ("<3 h light activity/week and no vigorous activity"), moderate (" $\geq 3$  h light activity/week or <1 h vigorous activity/week"), or high (" $\geq 1$  h vigorous activity/week").

https://doi.org/10.1371/journal.pmed.1003413.t001

of BSIs) deaths attributed to BSI. Participants who contracted a BSI or died from BSI were older, were more likely to be male and to be smokers, were less physically active and less educated, and had higher prevalence of self-reported history of cancer at study enrollment (Table 1).

The GRS explained 4.2% of the variation of BMI in our population (*F*-statistic = 2,461). There was a strong positive association between genetically predicted BMI and BSI incidence and mortality in the general population (Figs 1 and 2). A genetically predicted BMI of 20 kg/m<sup>2</sup>, 30 kg/m<sup>2</sup>, and 35 kg/m<sup>2</sup>—compared with 25 kg/m<sup>2</sup>—was associated with a hazard ratio (HR) for BSI incidence of 0.99 (95% confidence interval [CI]: 0.69 to 1.42; p = 0.960), 1.78 (95% CI: 1.40 to 2.27; p < 0.001), and 3.60 (95% CI: 2.18 to 5.93; p < 0.001), and for BSI mortality of 0.68 (95% CI: 0.48 to 0.96; p = 0.029), 2.56 (95% CI: 1.31 to 4.99; p = 0.006), and 8.57 (95% CI: 2.26 to 32.53; p = 0.002), respectively. Fig 1 and Fig 2 show tendencies of J-shaped associations, but are also compatible with linear relationships. A 1-kg/m<sup>2</sup> increase in genetically predicted BMI increased the risk of contracting BSI by 10% (95% CI: 5% to 16%; p < 0.001) and the risk of dying from BSI by 19% (95% CI: 6% to 33%; p = 0.003).

In the IPW analysis restricted to patients with BSI, we observed that a  $1 \cdot \text{kg/m}^2$  increase in genetically predicted BMI was associated with a non-statistically significant increased risk of BSI mortality of 11% (95% CI: -1% to 24%; p = 0.069). Genetically predicted BMI of 30 kg/m<sup>2</sup>,



Fig 1. Mendelian randomization analysis of body mass index and bloodstream infection incidence. The association between genetically predicted body mass index and risk of contracting a bloodstream infection (BSI), with a body mass index of  $25 \text{ kg/m}^2$  as reference (red dot). Gray lines represent 95% confidence intervals.

https://doi.org/10.1371/journal.pmed.1003413.g001

compared with 25 kg/m<sup>2</sup>, was associated with a HR for mortality of 2.34 (95% CI: 1.11 to 4.94; p = 0.025) (Fig 3). For comparison, a 1-kg/m<sup>2</sup> increase in BMI in the non-IPW MR analysis was associated with an increase in BSI mortality risk of 12% (95% CI: 1% to 25%; p = 0.039), and in the non-IPW traditional multivariable Cox regression model, it was associated with a non-statistically significant increase in mortality risk of 2% (95% CI: -1% to 5%; p = 0.188).

The sex-stratified associations between BMI and BSI incidence and mortality in the general population are presented in <u>S1</u> and <u>S2</u> Figs, respectively.

#### Sensitivity analyses

The traditional Cox regression model provided a weaker association between BMI and BSI incidence and mortality in the general population compared with the MR analysis (S1 and S2 Tables).

The results were unchanged after adjustment for 5 ancestry-informative principal components; in other words, there were no signs of population stratification (S3 Fig). When using only SNPs associated with the *FTO* gene to calculate the GRS, which explained 0.4% of BMI variation (*F*-statistic = 218), the results were consistent, though with wider CIs (S4 Fig).

Potential confounders were tabulated against BMI quartiles (S3 Table) and GRS quartiles (S4 Table). BMI was positively associated with self-reported history of cancer, never-smoking status, and age, and negatively associated with current smoking status, education, and physical activity. For the GRS, there was a positive association with current smoking status, while the other confounders were minimally associated ( $R^2 < 0.1\%$ ). Sensitivity analyses of never-smokers and ever-smokers yielded the same association between BMI and BSI mortality as in the main analysis (S5 Fig).

MR-Egger regression supported a causal association between BMI and BSI mortality in the general population adjusted for directional pleiotropy (HR = 1.18; 95% CI: 1.04 to 1.33; p = 0.011), and the average pleiotropic effect of the SNPs was null (HR = 1.00; 95% CI: 0.99 to 1.00; p = 0.476) (S5 Table; S6 Fig) [34]. We observed similar findings using the IVW method and weighted median estimator (p = 0.002 and p = 0.081, respectively). The comparable analyses in a 2-sample design yielded a similar association between BMI and risk of dying from BSI in the general population: The odds ratios in MR-Egger regression, IVW regression, and weighted median estimator analysis for a one standard deviation increase of BMI were 1.98 (95% CI: 0.95 to 4.18; p = 0.070), 1.89 (95% CI: 1.33 to 2.67; p < 0.001), and 2.09 (95% CI: 1.10 to 3.97; p = 0.025), respectively (S5 Table).

Evaluating signs of selection bias, we found that BSI patients in the highest BMI quartile, compared with those in the lowest quartile, were less likely to report a history of cancer, which was not the case in the general population (S6 Table). When we examined participants who would eventually develop BSI, participants in the highest BMI quartile—relative to those in the lowest BMI quartile—had a lower prevalence of risk factors and a higher prevalence of protective factors, compared with the concordant prevalences in the general population.

Among the underweight and normal weight participants (BMI  $< 25 \text{ kg/m}^2$ ), genetically predicted BMI was more strongly associated with BSI mortality in the general population for gram-positive infections than gram-negative infections (S7 Fig), but there were no differences among the overweight and obese participants.



Fig 2. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population. The association between genetically predicted body mass index and risk of dying from a bloodstream infection (BSI) in the general population, with a body mass index of 25 kg/m<sup>2</sup> as reference (red dot). Gray lines represent 95% confidence intervals.

https://doi.org/10.1371/journal.pmed.1003413.g002

#### Discussion

This MR study shows a positive association between genetically predicted BMI and BSI mortality, both in the general population and among patients with BSI, opposing the obesity paradox. Similarly, genetically predicted BMI was positively associated with BSI incidence.

In several studies, overweight and obese participants have been found to be at an increased risk of developing sepsis or BSI [7–10]. One of these studies also performed MR analyses of BMI and population risk of developing sepsis, but found no association, which may be due to the use of a weak genetic instrument [10]. Three recent population-based studies found overweight [7] and obese [7,9] participants to be at increased risk of BSI mortality, or observed a positive association between BMI and BSI mortality [11]. However, these observational studies may be affected by reverse causation and residual confounding.

The present study used a similar study population to that of Paulsen et al. [9], but had 6 more years of follow-up, 9,000 fewer participants (due to dependence on genotyping), and 55 (14%) more deaths due to BSI [9]. While Paulsen et al. found that BSI mortality risk started to increase at around BMI 30 kg/m<sup>2</sup>, we found it to start to increase at around 20 kg/m<sup>2</sup>. This observed difference is likely because the MR design has accounted for residual negative confounding. Our study found a stronger association between BMI and BSI mortality than the 2 other population-based studies [7,11]. MR estimates represent the impact of long-term differences in the trajectory of a risk factor over the life course; this impact is likely to differ from the impact of short-term interventions on the risk factor. Additionally, with a binary outcome, the estimates are marginal across covariates, and hence will differ from estimates from a multivariable-adjusted regression [36].

Our findings are in line with experimental animal studies of obesity and sepsis mortality [37]. However, human observational studies restricted to patients with sepsis or BSI have consistently found that there is a *reduced* risk of sepsis or BSI mortality among overweight and obese patients [18]. A systematic review from 2017 found markedly reduced mortality from sepsis among overweight and obese patients compared with those of normal weight [18]. Other hospital-based studies have reported similar findings [12–15], including a recent study of approximately 55,000 patients from 139 hospitals in the United States [17].

However, all of these hospital-based studies are likely affected by selection bias, in addition to possible reverse causation and confounding [38,39]. Similarly, when we restricted our study population to those that would develop BSI, the risk of BSI mortality among participants in the highest BMI quartile was reduced. Adjustment for introduced selection bias will often be insufficient [40].

One way to reduce selection bias is to use IPW [32]. In our IPW analysis of BSI patients, we observed a positive association between genetically predicted BMI and mortality similar to that of the total study population, albeit somewhat weaker. The finding of the comparable unweighted MR analysis was similar to that of the weighted one, which is supported by simulations that show that moderate selection bias will only lead to slight bias in a typical MR analysis [32]. Additionally, we observed a stronger association between BMI (both genetically predicted and observed) and BSI mortality in the general population than between BMI and BSI incidence, which would be expected if BMI increases BSI mortality risk among BSI patients.

In terms of the MR assumptions, assumption 1 was met as there was a strong association between the GRS and the exposure. Smoking was unevenly distributed at different levels of the





bloodstream infection, with a body mass index of 25 kg/m<sup>2</sup> as reference (red dot). The analysis was weighted for the inverse probability of contracting a bloodstream infection. Gray lines represent 95% confidence intervals.

https://doi.org/10.1371/journal.pmed.1003413.g003

GRS, potentially violating assumption 3. However, in analyses stratified by smoking status, we observed the same finding as in our main analysis. Additionally, assumption 2 was strengthened by MR-Egger regression, IVW regression, median estimator analysis, and the analysis restricted to the *FTO* gene as instrument. In the 2-sample sensitivity analysis, the linear association between BMI and risk of dying from a BSI in the general population was similar, albeit with less precision, compared with the 1-sample analyses. The larger standard errors suggest heterogeneity between the populations used by Yengo et al. [29] and in this study.

#### Strengths and limitations

To our knowledge, this is the first study to date to evaluate the association between BMI and risk of dying from BSI or sepsis using an MR design. This allows us to assess an association that is much less susceptible to reverse causation and unmeasured confounding than traditional observational studies. The MR design and IPW based on pre-disease background characteristics in the analysis of BSI mortality among BSI patients reduces the risk of selection bias. The genetic instrument for BMI that we used has been successfully applied in other recently published MR studies [41,42]. Importantly, although it is recommended that researchers create the genetic instrument based on the full set of SNPs associated with the desired exposure, this may increase the risk of pleiotropic effects and affect causal inference [24]. We therefore conducted a wide range of suggested sensitivity analyses [24], which supported that our findings were not meaningfully affected by pleiotropy. MR-Egger regression may be biased and affected by residual confounding when applied in a 1-sample setting [43]; however, in our 2-sample analyses we observed consistent results across the different sensitivity analyses. As the instruments were selected from an independent sample, our MR analyses will tend not to be affected by winner's curse [44]. We assumed the phenotype-disease association to be log-linear, which may bias the instrumental variable estimate depending on the properties of unmeasured confounders [45]; the non-instrumental variable analyses yielded similar but weaker associations. Given that we used a homogenous population of participants of northern European ancestry, the risk of population stratification is reduced, and it increases the precision of the GRS, which was based on a genome-wide association study of individuals of similar ancestry [29]. We encourage replication of our study in non-European cohorts to evaluate the generalizability of our findings to other populations. The vast majority of patients with BSI have sepsis, but not all patients with sepsis will have a positive blood culture [46]. Through record linkage with all hospitals in the region of interest, we had complete information on all relevant BSI occurrences. However, while blood culture tests are the gold standard to diagnose BSI, the sensitivity ranges between 10% and 50%, which may have reduced the statistical power in our study [47]. We overcame the issue of power by following approximately 56,000 participants representative of the adult Norwegian population for 23 years. Finally, we were able to allow for potential nonlinear associations between the exposure and the outcome, which has previously been challenging in MR studies [30].

#### Conclusion

In this Norwegian population-based cohort using an MR design, we found that any increase in genetically predicted BMI increased the risk of both BSI incidence and BSI mortality in the general population. Importantly, in an IPW MR analysis additionally accounting for selection bias, we found that increased genetically predicted BMI increased the risk of mortality also

among patients with BSI, opposing the obesity paradox. While other researchers have suggested that the "protective" effect of high BMI among patients with severe infectious diseases may help inform new therapeutic strategies [17,18], our findings do not encourage such initiatives. Given the steady, worldwide increase of mean BMI among adults, and that BSI is a common cause of death, these findings strongly support interventions to reduce the prevalence of overweight and obesity [2,5]. The efficacy of such interventions on the incidence of and mortality from BSI should then be studied.

#### **Supporting information**

**S1 Checklist. STROBE-MR checklist for Mendelian randomization studies.** (PDF)

**S2 Checklist. STROBE checklist for cohort studies.** (PDF)

S1 Fig. Mendelian randomization analysis of body mass index and bloodstream infection incidence, stratified by sex. The association between genetically predicted body mass index and risk of contracting a bloodstream infection among men (left panel) and women (right panel), with a body mass index of  $25 \text{ kg/m}^2$  as reference. Gray lines represent 95% confidence intervals.

```
(PDF)
```

**S2 Fig. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population, stratified by sex.** The association between genetically predicted body mass index and risk of dying from a bloodstream infection among men (left panel) and women (right panel) in the general population, with a body mass index of 25 kg/m<sup>2</sup> as reference. Gray lines represent 95% confidence intervals. (PDF)

**S3 Fig. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population, adjusted for 5 principal components.** The association between genetically predicted body mass index and risk of dying from a bloodstream infection in the general population, with a body mass index of 25 kg/m<sup>2</sup> as reference, adjusting for 5 principal components. Gray lines represent 95% confidence intervals. (PDF)

**S4 Fig. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population, with an instrument based on the** *FTO* gene. The association between genetically predicted body mass index and risk of dying from a bloodstream infection in the general population, with a body mass index of 25 kg/m<sup>2</sup> as reference. Gray lines represent 95% confidence intervals. Single nucleotide polymorphisms used were rs11075986, rs16952479, rs2075205, rs3751813, rs8047395, rs9922708, and rs9931164. (PDF)

**S5 Fig. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population, stratified by smoking status.** The association between genetically predicted body mass index and risk of dying from a bloodstream infection among never-smokers (left panel) and ever-smokers (right panel) in the general population, with a body mass index of 25 kg/m<sup>2</sup> as reference. Gray lines represent 95% confidence intervals. (PDF)

**S6 Fig. MR-Egger plot of bloodstream infection mortality in the general population.** Single nucleotide polymorphism–body mass index association and single nucleotide polymorphism–bloodstream infection mortality association in the general population, with inverse-variance-weighted regression for comparison.

(JPG)

**S7 Fig. Mendelian randomization analysis of body mass index and bloodstream infection mortality in the general population, stratified by gram-positive and gram-negative bacteria.** The association between genetically predicted body mass index and risk of dying from a bloodstream infection due to gram-positive (left panel) and gram-negative bacteria (right panel), with a body mass index of 25 kg/m<sup>2</sup> as reference. Gray lines represent 95% confidence intervals. There were 220 deaths due to gram-positive bloodstream infection and 215 deaths due to gram-negative bloodstream infection.

(PDF)

**S1 Table. Two-stage least-squares Mendelian randomization versus multivariable Cox regression of body mass index and bloodstream infection incidence.** BMI, body mass index; BSI, bloodstream infection; HR, hazard ratio. The 6 strata correspond to the BMI categories underweight, normal weight, overweight, obese class I, obese class II, and obese class III (from the top). The BMI strata were created based on residual BMI and observed BMI in the Mendelian randomization and multivariable analysis, respectively. The HR corresponds to a 1-kg/m<sup>2</sup> increase in BMI within that stratum.



**S2 Table. Two-stage least-squares Mendelian randomization versus multivariable Cox regression of body mass index and bloodstream infection mortality in the general population.** BMI, body mass index; BSI, bloodstream infection; HR, hazard ratio; INF, infinity. The 6 strata correspond to the BMI categories underweight, normal weight, overweight, obese class I, obese class II, and obese class III (from the top). The BMI strata were created based on residual BMI and observed BMI in the Mendelian randomization and multivariable analysis, respectively. The HR corresponds to a 1-kg/m<sup>2</sup> increase in BMI within that stratum. (DOCX)

S3 Table. Distribution of potential confounders by body mass index quartiles. BMI, body mass index; Q, quartile; SD, standard deviation. Post-secondary defined as at least some university or other post-secondary education. Moderate/high activity defined as  $\geq$ 3 h light activity/week or any vigorous activity/week.

(DOCX)

S4 Table. Distribution of potential confounders by genetic risk score for body mass index quartiles. BMI, body mass index; GRS, genetic risk score; Q, quartile; SD, standard deviation. Post-secondary defined as at least some university or other post-secondary education. Moderate/high activity defined as  $\geq$ 3 h light activity/week or any vigorous activity/week. (DOCX)

S5 Table. Mendelian randomization sensitivity analyses of linear association between body mass index and bloodstream infection mortality in the general population. HR, hazard ratio; IVW, inverse-variance-weighted; OR, odds ratio. The analyses assume a linear relation-ship between body mass index and bloodstream infection mortality in the general population using the same 939 single nucleotide polymorphisms (SNPs) as used to create the genetic risk score. Two-sample analyses use SNP–exposure associations from Yengo et al. [29] and SNP– outcome associations from HUNT. The  $I^2$  values of the SNP–exposure associations were 54%

in the 1-sample MR-Egger regression and 92% in the 2-sample MR-Egger regression. Effect estimates reported as HR for 1 unit increase of body mass index in 1-sample analyses and as OR for 1 standard deviation increase of body mass index in 2-sample analyses. (DOCX)

S6 Table. Distribution of potential confounders by body mass index quartiles among patients with bloodstream infection. BMI, body mass index; BSI, bloodstream infection; Q, quartile; SD, standard deviation. The final column, for dichotomous covariates, compares the Q4/Q1 ratio among patients with BSI with the Q4/Q1 ratio in the general population (S3 Table), and similarly compares the mean difference of Q4 – Q1 among BSI participants with Q4 – Q1 in the general population. Post-secondary defined as at least some university or other post-secondary education. Moderate/high activity defined as  $\geq$ 3 h light activity/week or any vigorous activity/week.

(DOCX)

S1 Text. Supplementary description of methods.

(DOCX)

#### Acknowledgments

HUNT is a collaboration between the HUNT Research Centre (Faculty of Medicine and Health Sciences, Norwegian University of Science and Technology), Nord-Trøndelag County Council, Central Norway Regional Health Authority, and Norwegian Institute of Public Health.

This paper does not necessarily reflect the position or policy of the US Government or Department of Veterans Affairs.

#### **Author Contributions**

- **Conceptualization:** Tormod Rogne, Erik Solligård, Stephen Burgess, Bjørn O. Åsvold, Andrew T. DeWan, Jan K. Damås.
- **Data curation:** Tormod Rogne, Erik Solligård, Ben M. Brumpton, Julie Paulsen, Randi M. Mohus, Arne Mehl, Jan K. Damås.

Formal analysis: Tormod Rogne, Stephen Burgess, Ben M. Brumpton, Bjørn O. Åsvold.

Funding acquisition: Erik Solligård, Andrew T. DeWan, Jan K. Damås.

Investigation: Stephen Burgess, Randi M. Mohus, Bjørn O. Åsvold.

Methodology: Tormod Rogne, Stephen Burgess, Ben M. Brumpton, Bjørn O. Åsvold, Andrew T. DeWan.

**Project administration:** Erik Solligård, Ben M. Brumpton, Lise T. Gustad, Arne Mehl, Andrew T. DeWan, Jan K. Damås.

Resources: Erik Solligård, Andrew T. DeWan, Jan K. Damås.

Software: Erik Solligård.

Supervision: Erik Solligård, Stephen Burgess, Andrew T. DeWan, Jan K. Damås.

Validation: Erik Solligård, Hallie C. Prescott.

Visualization: Stephen Burgess.

- Writing original draft: Tormod Rogne, Erik Solligård, Stephen Burgess, Bjørn O. Åsvold, Andrew T. DeWan, Jan K. Damås.
- Writing review & editing: Tormod Rogne, Erik Solligård, Stephen Burgess, Ben M. Brumpton, Julie Paulsen, Hallie C. Prescott, Randi M. Mohus, Lise T. Gustad, Arne Mehl, Bjørn O. Åsvold, Andrew T. DeWan, Jan K. Damås.

#### References

- Singer M, Deutschman CS, Seymour CW, Shankar-Hari M, Annane D, Bauer M, et al. The third international consensus definitions for sepsis and septic shock (Sepsis-3). JAMA. 2016; 315(8):801–10. https://doi.org/10.1001/jama.2016.0287 PMID: 26903338
- Goto M, Al-Hasan M. Overall burden of bloodstream infection and nosocomial bloodstream infection in North America and Europe. Clin Microbiol Infect. 2013; 19(6):501–9. <u>https://doi.org/10.1111/1469-0691.12195</u> PMID: 23473333
- Fleischmann C, Scherag A, Adhikari NKJ, Hartog CS, Tsaganos T, Schlattmann P, et al. Assessment of global incidence and mortality of hospital-treated sepsis current estimates and limitations. Am J Respir Crit Care Med. 2016; 193(3):259–72. https://doi.org/10.1164/rccm.201504-0781OC PMID: 26414292
- Dat VQ, Vu HN, Nguyen The H, Nguyen HT, Hoang LB, Vu Tien Viet D, et al. Bacterial bloodstream infections in a tertiary infectious diseases hospital in Northern Vietnam: aetiology, drug resistance, and treatment outcome. BMC Infect Dis. 2017; 17(1):493. <u>https://doi.org/10.1186/s12879-017-2582-7</u> PMID: 28701159
- NCD Risk Factor Collaboration. Rising rural body-mass index is the main driver of the global obesity epidemic in adults. Nature. 2019; 569(7755):260–4. https://doi.org/10.1038/s41586-019-1171-x PMID: 31068725
- Di Angelantonio E, Bhupathiraju SN, Wormser D, Gao P, Kaptoge S, de Gonzalez AB, et al. Body-mass index and all-cause mortality: individual-participant-data meta-analysis of 239 prospective studies in four continents. Lancet. 2016; 388(10046):776–86. https://doi.org/10.1016/S0140-6736(16)30175-1 PMID: 27423262
- Twig G, Geva N, Levine H, Derazne E, Goldberger N, Haklai Z, et al. Body mass index and infectious disease mortality in midlife in a cohort of 2.3 million adolescents. Int J Obes. 2018; 42(4):801–7.
- Wang HE, Griffin R, Judd S, Shapiro NI, Safford MM. Obesity and risk of sepsis: a population-based cohort study. Obesity. 2013; 21(12):762–9. https://doi.org/10.1002/oby.20468 PMID: 23526732
- Paulsen J, Askim Å, Mohus RM, Mehl A, Dewan A, Solligård E, et al. Associations of obesity and lifestyle with the risk and mortality of bloodstream infection in a general population: a 15-year follow-up of 64 027 individuals in the HUNT Study. Int J Epidemiol. 2017; 46(5):1573–81. https://doi.org/10.1093/ije/ dyx091 PMID: 28637260
- Winter-Jensen M, Afzal S, Jess T, Nordestgaard BG, Allin KH. Body mass index and risk of infections: a Mendelian randomization study of 101,447 individuals. Eur J Epidemiol. 2020; 35(4):347–54. <u>https://</u> doi.org/10.1007/s10654-020-00630-7 PMID: 32307655
- 11. Williams PT. Inadequate exercise as a risk factor for sepsis mortality. PLoS ONE. 2013; 8(12):e79344. https://doi.org/10.1371/journal.pone.0079344 PMID: 24324580
- Sasabuchi Y, Yasunaga H, Matsui H, Lefor AT, Horiguchi H, Fushimi K, et al. The dose-response relationship between body mass index and mortality in subjects admitted to the ICU with and without mechanical ventilation. Respir Care. 2015; 60(7):983–91. https://doi.org/10.4187/respcare.03694
  PMID: 25691767
- Nguyen AT, Tsai C, Hwang L, Lai D, Markham C. Obesity and mortality, length of stay and hospital cost among patients with sepsis: a nationwide inpatient retrospective cohort study. PLoS ONE. 2016; 11(4): e0154599. https://doi.org/10.1371/journal.pone.0154599 PMID: 27124716
- Zhou Q, Wang M, Li S, Zhang J, Ma Q, Ding Y, et al. Impact of body mass index on survival of medical patients with sepsis: a prospective cohort study in a university hospital in China. BMJ Open. 2018; 8(9): e021979. https://doi.org/10.1136/bmjopen-2018-021979 PMID: 30209156
- Atamna A, Elis A, Gilady E, Gitter-Azulay L, Bishara J. How obesity impacts outcomes of infectious diseases. Eur J Clin Microbiol Infect Dis. 2017; 36(3):585–91. <u>https://doi.org/10.1007/s10096-016-2835-1</u> PMID: 27864622
- Prescott HC, Chang VW, O'Brien JM, Langa KM, Iwashyna T. Obesity and one-year outcomes in older Americans with severe sepsis. Crit Care Med. 2014; 42(8):1766–74. <u>https://doi.org/10.1097/CCM.</u> 00000000000336 PMID: 24717466

- Pepper DJ, Demirkale CY, Sun J, Rhee C, Fram D, Eichacker P, et al. Does obesity protect against death in sepsis? A retrospective cohort study of 55,038 adult patients. Crit Care Med. 2019; 47(5):643– 50. https://doi.org/10.1097/CCM.0000000003692 PMID: 30789403
- Wang S, Liu X, Chen Q, Liu C, Huang C, Fang X. The role of increased body mass index in outcomes of sepsis: a systematic review and meta-analysis. BMC Anesthesiol. 2017; 17(1):118. https://doi.org/10. 1186/s12871-017-0405-4 PMID: 28859605
- Carnethon MR, DeChavez PJD, Biggs MI, Lewis CE, Pankow JS, Bertoni AG. Association of weight status with mortality in adults with incident diabetes. JAMA. 2012; 308(6):581–90. <u>https://doi.org/10.1001/jama.2012.9282</u> PMID: 22871870
- 20. Banack HR, Kaufman JS. The "obesity paradox" explained. Epidemiology. 2013; 24(3):461–2. https:// doi.org/10.1097/EDE.0b013e31828c776c PMID: 23549182
- Ng PY, Eikermann M. The obesity conundrum in sepsis. BMC Anesthesiol. 2017; 17(1):147. <u>https://doi.org/10.1186/s12871-017-0434-z PMID: 29070011</u>
- Goulden R. Alternative explanations for the obesity paradox in sepsis. Crit Care Med. 2019; 47(9):e790. https://doi.org/10.1097/CCM.00000000003844 PMID: 31415325
- Lajous M, Banack HR, Kaufman JS, Hernán MA. Should patients with chronic disease be told to gain weight? The obesity paradox and selection bias. Am J Med. 2015; 128(4):334–6. <u>https://doi.org/10.1016/j.amjmed.2014.10.043</u> PMID: 25460531
- 24. Davies NM, Holmes MV, Davey Smith G. Reading Mendelian randomisation studies: a guide, glossary, and checklist for clinicians. BMJ. 2018; 362:k601. https://doi.org/10.1136/bmj.k601 PMID: 30002074
- Davey Smith G, Davies N, Dimou N, Egger M, Gallo V, Golub R, et al. STROBE-MR: guidelines for strengthening the reporting of Mendelian randomization studies. PeerJ Prepr. 2019; 7:e27857v1. https://doi.org/10.7287/peerj.preprints.27857v1
- von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. Lancet. 2007; 370(9596):1453–7. <u>https://doi.org/10.1016/S0140-6736(07)</u> 61602-X PMID: 18064739
- Holmen J, Midthjell K, Krüger Ø, Langhammer A, Holmen TL, Bratberg GH, et al. The Nord-Trøndelag Health Study 1995–97 (HUNT 2): objectives, contents, methods and participation. Nor Epidemiol. 2003; 13(1):19–32.
- Pien BC, Sundaram P, Raoof N, Costa SF, Mirrett S, Woods CW, et al. The clinical and prognostic importance of positive blood cultures in adults. Am J Med. 2010; 123(9):819–28. <u>https://doi.org/10.1016/j.amjmed.2010.03.021</u> PMID: 20800151
- 29. Yengo L, Sidorenko J, Kemper KE, Zheng Z, Wood AR, Weedon MN, et al. Meta-analysis of genomewide association studies for height and body mass index in ~ 700000 individuals of European ancestry. Hum Mol Genet. 2018; 27(20):3641–9. https://doi.org/10.1093/hmg/ddy271 PMID: 30124842
- Sun Y-Q, Burgess S, Staley JR, Wood AM, Bell S, Kaptoge SK, et al. Body mass index and all cause mortality in HUNT and UK Biobank studies: linear and non-linear Mendelian randomisation analyses. BMJ. 2019; 364:I1042. https://doi.org/10.1136/bmj.I1042 PMID: 30957776
- Palmer TM, Sterne JAC, Harbord RM, Lawlor DA, Sheehan NA, Meng S, et al. Instrumental variable estimation of causal risk ratios and causal odds ratios in Mendelian randomization analyses. Am J Epidemiol. 2011; 173(12):1392–403. https://doi.org/10.1093/aje/kwr026 PMID: 21555716
- Gkatzionis A, Burgess S. Contextualizing selection bias in Mendelian randomization: how bad is it likely to be? Int J Epidemiol. 2019; 48(3):691–701. https://doi.org/10.1093/ije/dyy202 PMID: 30325422
- Dong SS, Zhang YJ, Chen YX, Yao S, Hao RH, Rong Y, et al. Comprehensive review and annotation of susceptibility SNPs associated with obesity-related traits. Obes Rev. 2018; 19(7):917–30.
- Burgess S, Thompson SG. Interpreting findings from Mendelian randomization using the MR-Egger method. Eur J Epidemiol. 2017; 32(5):377–89.
- Wang C, Zhan X, Liang L, Abecasis GR, Lin X. Improved ancestry estimation for both genotyping and sequencing data using projection procrustes analysis and genotype imputation. Am J Hum Genet. 2015; 96(6):926–37. https://doi.org/10.1016/j.ajhg.2015.04.018 PMID: 26027497
- 36. Burgess S. Consistency and collapsibility are they crucial for instrumental variable analysis with a survival outcome in mendelian randomization? Epidemiology. 2015; 26(3):411–3. https://doi.org/10.1097/ EDE.00000000000282 PMID: 25835137
- Mittwede PN, Clemmer JS, Bergin PF, Xiang L. Obesity and critical illness. Shock. 2016; 45(4):349–58. https://doi.org/10.1097/SHK.0000000000512 PMID: 26513706
- Munafò MR, Tilling K, Taylor AE, Evans DM, Smith GD. Collider scope: when selection bias can substantially influence observed associations. Int J Epidemiol. 2018; 47(1):226–35. <u>https://doi.org/10.1093/</u> ije/dyx206 PMID: 29040562

- Arabi YM, Dara SI, Tamim HM, Rishu AH, Bouchama A, Khedr MK, et al. Clinical characteristics, sepsis interventions and outcomes in the obese patients with septic shock: an international multicenter cohort study. Crit Care. 2013; 17(2):R72. https://doi.org/10.1186/cc12680 PMID: 23594407
- Hernán MA, Hernández-Díaz S, Robins JM. A structural approach to selection bias. Epidemiology. 2004; 15(5):615–25. https://doi.org/10.1097/01.ede.0000135174.63482.43 PMID: 15308962
- Takahashi H, Cornish AJ, Sud A, Law PJ, Disney-Hogg L, Calvocoressi L, et al. Mendelian randomization provides support for obesity as a risk factor for meningioma. Sci Rep. 2019; 9(1):309. <u>https://doi.org/10.1038/s41598-018-36186-6</u> PMID: 30670737
- 42. Cheung CL, Tan KCB, Au PCM, Li GHY, Cheung BMY. Evaluation of GDF15 as a therapeutic target of cardiometabolic diseases in human: a Mendelian randomization study. EBioMedicine. 2019; 41:85–90. https://doi.org/10.1016/j.ebiom.2019.02.021 PMID: 30772304
- **43.** Minelli C, Fabiola Del Greco M, Van der Plaat D, Bowden J, Sheehan N, Thompson J. The use of twosample methods for Mendelian randomization analyses on single large datasets. bioRxiv. 2020 May 7. https://doi.org/10.1101/2020.05.07.082206
- Taylor AE, Davies NM, Ware JJ, Vanderweele T, Smith GD, Munafò MR. Mendelian randomization in health research: using appropriate genetic variants and avoiding biased estimates. Econ Hum Biol. 2014; 13(1):99–106. https://doi.org/10.1016/j.ehb.2013.12.002 PMID: 24388127
- Palmer TM, Thompson JR, Tobin MD, Sheehan NA, Burton PR. Adjusting for bias and unmeasured confounding in Mendelian randomization studies with binary responses. Int J Epidemiol. 2008; 37 (5):1161–8. https://doi.org/10.1093/ije/dyn080 PMID: 18463132
- 46. Torsvik M, Gustad LT, Mehl A, Bangstad IL, Vinje LJ, Damås JK, et al. Early identification of sepsis in hospital inpatients by ward nurses increases 30-day survival. Crit Care. 2016; 20(1):244. <u>https://doi.org/ 10.1186/s13054-016-1423-1 PMID: 27492089</u>
- Nguyen MH, Clancy CJ, Pasculle AW, Pappas PG, Alangaden G, Pankey GA, et al. Performance of the T2bacteria panel for diagnosing bloodstream infections. Ann Intern Med. 2019; 170(12):845–52. <u>https:// doi.org/10.7326/M18-2772 PMID: 31083728</u>